Author’s response to reviews

Title: PRESERVATION OF RENAL FUNCTION IN CARDIAC SURGERY PATIENTS WITH LOW CARDIAC OUTPUT SYNDROME: LEVOSIMENDAN VS BETA AGONISTS

Authors:

JOSE LUIS GUERRERO ORRIACH (guerreroorriach@gmail.com)

ISABEL Arce (isanavarroarce@gmail.com)

PAULA RODRIGUEZ (paulahernandez81@gmail.com)

aida Ponferrada (aidaraigonp@gmail.com)

ALFREDO Manso (alfredomalomanso@gmail.com)

MARTA ALIAGA (martaram85@gmail.com)

ALICIA FERNANDEZ (aliramfdez@gmail.com)

JUAN JOSE BELMONTE (jjescalonabelmonte@gmail.com)

INMACULADA ESTEVEZ (inmabellido@gmail.com)

AURELIO LUQUE (jagomez@uma.es)

RAFAEL SERRANO (rafbaser@gmail.com)

CARMEN MEDINA (carmentm_86@hotmail.com)

MANUEL navarro (mrubionav@gmail.com)

jose mañas (jose.cruz.sspa@juntadeandalucia.es)

Version: 8 Date: 10 Oct 2019

Author’s response to reviews:

Editor Comments:

1. Ethics approval

   -- Please confirm that your study was both submitted to and approved by the ethics committee in this "ethics approval and consent to participate" section.
We have included:

Our study was both submitted to and approved by the ethics committee.

The study was approved by the local Ethics Committee (Malaga Norte- IBIMA). Patients were divided into two groups based on the therapy received.

2. Consent to participate

-- Please also include the consent to participate information seen in your methods section in the "ethics approval and consent to participate" section .

We have included:

Informed consent to participate in the study was obtained from participants.